
Great Basin Incident Commander Conference Call Agenda 2016 

 
Date:      Saturday, August 06, 2016 Note taker:  Gina Bald 
GB MAC/GACC Coordinator:   
   
Agenda/Incident Commander Discussion Points: Summarize the following in 4 min or less: 
• Fire Size / % contained / # of people / Cost to date 
• Significant changes since the last 209 
• BRIEF fire synopsis /Critical Incident Information (use Directional vs DIVS’s) 
• Political Issues / Community Meetings 
• Closures / Evacuations / Public Safety Issues 
• Major Safety Concerns / Injuries / Accidents 
• Critical Resource Needs 
• Resources Available to Reassign/Share/Demob 
 
Predictive Services: 

Weather update: 
Basil Newmerzhycky 

Some lighting late yesterday and overnight in Central and Eastern NV 

towards S. UT.   Primarily wet.  Lightning in the mountains of Central ID, E. 

OR, Northern Payette NF, Boise NF, ID and WY border.  Today:  Lightning in 

E. ID, W. WY. Gusty winds – front approaching.  50 mph downdrafts.  NV 

drier air – gusty in the south 2/3 of NV.  Shower activity the past 2-3 days in 

these areas. Sunday:  Strong cold front in ID and UT, WY.  Lighting on the 

Payette, Boise and part of Salmon-Challis NFs.  Wetting rains on the Bridger-

Teton NF.  Winds in UT.  Drier air early next week. 
GBCC Update: 
Gina Dingman 

• Minimal IA.  4 new starts for .4 acres 
• Large Fire Growth: 8532 
• 2 MAFFS in BOI, 5 ATs, 1 VLAT, 21 SEATs (2 day off), 3 LPs (2 

more available at 1630), 4 loads of Smokejumpers at 4 bases 
(OGD, MYL, PIH, WMC).  Construction at Ely airport – moved 
Jumpers to WMC.  

• Reassigning resources form Poodle and Virginia Mtn Complex 
• Staging Areas in SLC, TWF and, EKO are at capacity 

 
1 
Fire: Pioneer 
IC: Beth Lund 
 

58136 / 20% /1500 /24M  
• Increase of 8000 acres 
• Exciting day yesterday. Thunder cell sat over NE portion of fire.   

Some moisture last night. Downdrafts blew it over Rock Creek 
area. Jumped Payette River.  3000 acres over the river. Not as bad 
as expected.  Difficult with predicted weather today.  Changed 
strategy.  Changed Divisions and Branches. 

• West and South:  Holding 
• East:  Modules putting in contingency line. No significant growth.   



• Evacuation went smoothly. Escorted down Hwy 21. Hazard trees 
removed.  Hwy 17 – Powepoles burned, rock debris – unstable 
route in.  Lowman is out of power. 

• Critical Resource Needs:  2 T1 IHC 
• Political issues:  Keeping up with social media and Lowman 

evacuation 
• Closures:  Routes closed off. Expedite opening Hwy 21. 
• Injuries:  None 
• Resources available for release: None.  Resources coming up on 

14th day. 
2 
Fire:  Broad Mouth 
IC:   
Report:  Snider 

20614 / 35% / 29% / 205 
• Finance just got in place, so not comfortable with giving a cost to 

date 
• North:  ID Border.  Hand checking. 
• East:  IR showed 2 spots of heat.  Fence posts – will put out today. 
• South:  905 solid.  Some heat.  Hold today. 
• West:  Most work.  Indirect dozer line.  A lot of heat.  Crews 

working here. 
• Injuries:  1 minor – barbed wire fence 
• Political Issues:  Media interest.  International Long Range 

Competition starting on West Flank.  They are working well with 
the IMT.  Have placed at every stand a firefighter, Safety Officer, 
Rangemaster and Shooter. They do have an evacuation plan. 

• Critical Resource Needs:  None 
• Resources available for Release:  T2 Crew 

3 
Fire: Box Canyon 
IC: Pilgrim 

2313 / 0% / 84 / $613500 
• Changes since last 209 reported:  110 acreage gain 
• 70% of incident is in monitor and patrol status; 30 % taking 

action. (0% contained) 
• Weather calm. No rain received. 
• Burning towards Hell’s Kitchen basin.  Heavy helos working here.  

Steep, rocky, spotting.  Spread south to Ericson Basin – spotting.  
Will monitor today. 

• Potential to burn into 2 MAP areas. 
• Political Issues:  Heavy recreation are, reservoir. 
• Closures:  2 Campgrounds.  Closure postponed until next week.  

Hiking trails and dirt roads closed. 
• No injuries 
• Two FS vehicles damaged by tractor trailer hitting a tree and tree 

fall damaging vehicles.  Driver left scene 
• Critical Resource Needs: T3 Helo (received)    
• Resources to assign:  1 T2 Crew t 

4 
Fire: Cliff Creek 
IC: Russ Bird 
Report:  Kuo 

31,966 acres, 88% contained, 268 personnel, $12.9mm 
 
• ICT3:  Paul Swenson with Jeff Liday as ICT3-T 
• Significant Changes since last 209: 300 acres 



 
 
 
 

• Pretty calm day yesterday. Everything holding.   
• West: Yesterday the evac order was lifted for residents on west 

side.  Re-entry went well, no issues.  Powerline re-energized.   
• South:  Patrol by air. 
• North:  Natural barriers in Gros Ventre and Schoal Wilderness Study 

Area. 
• East: Hold and secure eastern perimeter from rock ridge in 

Wilderness (north) to Jacks Creek (south).    
• Political issues/Community Meeting: None planned.   
• Closures: Forest closures modified yesterday to allow access into 

forest west of the fire and north of Hwy.  Day use only.  However 
forest still closed to area impacted by fire.   

• Injury/Accident: None in last 24 hrs.   
• Critical Resource Needs: None.   
• Resources to reassign: None at this time.  On the glide path, scaling 

down.  Western WY Type 3 Team shadowing GB Team 5 today.  TOC 
planned for 0600 hrs Sunday.  

5 
Virginia Mts Complex 
IC: Tim Roide 

59727 / 100% /342  /7.9M   
• No significant changes to 209 Report 
• Me all control and suppression repair objectives. 
• Turned back to Unit at 0600 
• Political Issues:  Tribal Council meeting. Impacts to Sutcliff and 

Pyramid Reservation.  
• No Closures  
• No injuries 
• Critical Resource Needs:  None 
• No Resources  available to reassign 

 
6 
Fire:  Sheep Creek 
IC:  Armantrout 

1061 / 100% /150  /$850000 
• Transfer to T4 at 1800 
• Continue to mop-up and patrol 
• Some moisture, not significant 
• No political issues 
• Closures:  Lifted 
• Critical Resource Needs:  None 

7 
Poodle 
IC:  Uhlig 
 

6557 / 98% /104 /1.625 
• Back over to District today. 
• Last report 
• Braking down ICP tomorrow 
• Nothing else to report 
•  

MAC / GBCC 
Coordinator 
Derrek Hartman 

No Report 

Date/Time of Next Call:   8/7 @ 0800 MT / 0900 PT  
 


